Author's response to reviews

Title: Factors associated with maternal anaemia among pregnant women in Dhaka city

Authors:

Hasina Akhter Chowdhury (hachowdhury01@gmail.com)
Kazi Rumana Ahmed (lopa_birdem@yahoo.com)
Fatema Jebunnesa (runabd2005@yahoo.com)
Jesmin Akter (uttaron_007@yahoo.mail)
Sharmin Hossain (shormybd@yahoo.com)
Md Shahjahan (mdshahjahan@agnionline.com)

Version: 14 Date: 8 June 2015

Author's response to reviews:

Comments:
On the basis of the comments we have received we feel that your manuscript is suitable for publication in the journal. However, before we proceed with acceptance there are some further editorial revisions which we require you to make. Please note that your manuscript cannot be accepted for publication until these revisions have been satisfactorily addressed.

Editorial Requirement:

*Post-review copyediting: We recommend that you copyedit the paper to improve the style of written English.

Reply: One of the expert [Mr. M. Shamsul Islam Khan, Advisor, Department of Library and Part-Time Faculty, English Foundation Course, Bangladesh University of Health Sciences, Phone: 9010952, 8035501-6 (ext1313); email address: msik1949@yahoo.com] revised the manuscript according to your suggestions.

Comments:

Please also ensure that your revised manuscript conforms to the journal style (http://www.biomedcentral.com/info/ifora/medicine_journals ). It is important that your files are correctly formatted.

Reply: Revised the manuscript according to your journal format.